Author's response to reviews

Title: Variability in depression prevalence in early rheumatoid arthritis: a comparison of the CES-D and HAD-D Scales

Authors:

Tanya Covic (t.covic@uws.edu.au)
Julie F Pallant (jpallant@unimelb.edu.au)
Alan Tennant (a.tennant@leeds.ac.uk)
Sally Cox (S.Cox@leeds.ac.uk)
Paul Emery (p.emery@leeds.ac.uk)
Philip G Conaghan (P.Conaghan@leeds.ac.uk)

Version: 5 Date: 4 February 2009

Author's response to reviews:

Dear BMC,

Thanks you for the opportunity to reproof the document. We have made one minor change in the references, otherwise the paper is fine.